Author's response to reviews

Title: Developing theory-informed behaviour change interventions to implement evidence into practice: a systematic approach using the Theoretical Domains Framework

Authors:

Simon D French (s.french@unimelb.edu.au)
Sally E Green (sally.green@monash.edu)
Denise A O'Connor (denise.oconnor@monash.edu)
Joanne E McKenzie (joanne.mckenzie@monash.edu)
Jill J Francis (j.francis@abdn.ac.uk)
Susan Michie (s.michie@ucl.ac.uk)
Rachelle Buchbinder (rachelle.buchbinder@monash.edu)
Peter Schattner (peter.schattner@monash.edu)
Neil Spike (neil.spike@vma.com.au)
Jeremy M Grimshaw (jgrimshaw@ohri.ca)

Version: 4 Date: 3 February 2012

Author's response to reviews:

Dear Martin,

We have uploaded a final "clean" version of the manuscript after considering the copy-editor’s comments. We accepted most of the suggested changes, but rejected one suggestion - to change "health care" to "healthcare" (first sentence of the Background and in a paragraph on page 5, also Background section). My understanding is that "health care" is the noun, and "healthcare" is the adjective, whereas the copy editor has suggested changing all instances of this term to "healthcare". There was one instance where I accepted the suggested change on page 5 for "healthcare professional".

As previously discussed, I also added an Acknowledgement to you Martin for your assistance in the editorial process.

Kind regards

Simon French, on behalf of the IMPLEMENT author team